Author's response to reviews

Title: Age differences in mental health literacy

Authors:

Louise Farrer (louise.farrer@anu.edu.au)
Liana Leach (liana.leach@anu.edu.au)
Kathleen M Griffiths (kathy.griffiths@anu.edu.au)
Helen Christensen (helen.christensen@anu.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)

Version: 5 Date: 10 April 2008

Author's response to reviews:

Cover letter addressing reviewer comments

Referee 1: Julie Wetherell

Point 1: Suggestion to include a sentence at the end of the introduction describing the findings that might be expected, including the age cut-offs used.

The following sentence was added to the end of the introduction:

“Although this investigation is exploratory, based on the results of previous studies we anticipate that the oldest (70+ years) and youngest (18-24 years) respondents will report lower levels of recognition of depression and schizophrenia, respectively, than respondents in other age groups”. 